Author's response to reviews

Title: The impact of donor policies in Europe: a steady increase, but not everywhere

Authors:

Remco Coppen (r.coppen@nivel.nl)
Roland D Friele (r.friele@nivel.nl)
Sjef K M Gevers (j.k.gevers@amc.uva.nl)
Geke A Blok (geke.blok@erd.unimaas.nl)
Jouke Van der Zee (j.vanderzee@nivel.nl)

Version: 3 Date: 10 November 2008

Author's response to reviews:

Dear Editor,

Due to my holidays I was not able to respond to your request to submit the formatting changes for MS:5154525932120145 earlier.

As you requested, we have made the following formatting changes:

Major Revisions: The text Box 1 has been transformed to Table 1. To make the contents look more like a Table we have also made some layout adjustments.

Minor Revisions: We have cropped the figures according to your online figure guide.

Furthermore we have checked the manuscript for typological errors.

Yours sincerely,

Remco Coppen